Author's response to reviews

Title: Bicultural identity among economical migrants from three south European countries living in Switzerland: Adaptation and validation of a new psychometric instrument

Authors:

Ariel Eytan (Ariel.Eytan@hcuge.ch)
Nuria Jene-Petschen (Nuria.Jene-Petschen@hcuge.ch)
Marianne Gex-Fabry (Marianne.Gex-Fabry@hcuge.ch)

Version: 3 Date: 4 May 2007

Author's response to reviews:

To the BioMed Central Editorial Production Team:

We made the requested changes:

Formatting changes requested

Keywords and Title sheet heading. Please remove these.
Done

Please remove unnecessary block capitalisation of Author names.
Done

Qualifications - Please remove the authors' qualifications (e.g. PhD / MD) from the manuscript and submissions system.
Done

Manuscript sections - Manuscript sections should include (in the following order): Abstract; Background; (not Introduction) Discussion; Conclusions; Abbreviations (if any); Competing interests; Authors’ contributions; Acknowledgements; References*; Figure legends (if any); Tables (if any); Description of Additional files (if any).

*currently wrongly placed.
Done

Looking forward to see our paper published,

Best wishes,

Ariel Eytan.